Author's response to reviews

Title: Perceived barriers of heart failure nurses and cardiologists in using clinical decision support systems in the treatment of heart failure patients

Authors:

Arjen de Vries (a.e.de.vries@thorax.umcg.nl)
Martje van der Wal (m.h.l.van.der.wal@umcg.nl)
Maurice Nieuwenhuis (m.m.w.nieuwenhuis@umcg.nl)
Richard de Jong (richard.dejong@wza.nl)
Rene van Dijk (rene@cavari.nl)
Tiny Jaarsma (tiny.jaarsma@liu.se)
Hans Hillege (h.hillege@umcg.nl)
Rene Jorna (r.j.j.m.jorna@rug.nl)

Version: 3 Date: 30 March 2013

Author's response to reviews: see over
Groningen, 30 March 2013

Dear editor of the journal ‘BioMed Central’, Medical informatics and Decision making,

We thank the reviewers for their comments on our manuscript ‘Perceived barriers of heart failure nurses and cardiologists in using clinical decision support systems in the treatment of heart failure patients’. We looked carefully at all the comments and revised the paper as advised. We believe that the comments have helped us to improve and strengthen the manuscript. Modifications that were made are explained in a detailed revision report and the revisions made in the manuscript are underlined in the revised manuscript itself.

Enclosed you find the revised manuscript, including a detailed response to the reviewer’s comments.

On behalf of all my co-authors, we would like to resubmit this revised manuscript and hope that you are willing to accept our manuscript.

Sincerely, on behalf of Professor Hans H. Hillegie, Professor Tiny Jaarsma and Professor Rene Jorna

Arjen. E de Vries, MA
Department of Cardiology
University Medical Centre Groningen
PO Box 30.001
9700 RB Groningen
Email: a.e.de.vries@thorax.umcg.nl
The Netherlands